Author's response to reviews

Title: Pattern of neuropsychological performance among HIV positive patients in Uganda

Authors:

Kevin R Robertson (kevinr@neurology.unc.edu)
Noeline Nakasujja (drnoeline@yahoo.com)
Matthew Wong (matthewharrywong@hotmail.com)
Seggane Musisi (segganemusisi@yahoo.ca)
Ellie Katabira (katabira@infocom.co.ug)
Thomas D Parsons (TParsons@ict.usc.edu)
Allan Ronald (aronald@ms.umanitoba.ca)
Ned Sacktor (sacktor@jhmi.edu)

Version: 6  Date: 4 April 2007

Author's response to reviews: see over
Dear Editors:

Please accept the requested additional formatting changes to the manuscript entitled ‘Pattern of neuropsychological performance among HIV positive patients in Uganda’. Please see an itemized list of our revisions below. We want to thank the editors and reviewers for their work in making the manuscript much improved.

Sincerely,

Kevin Robertson, Ph.D.
Professor, Director of Neuropsychology
Neurology, School of Medicine
University of North Carolina at Chapel Hill

Formatting changes requested:

1. Country - Please remember to include the country in the affiliation details.
   The country was added to all affiliations.

2. Please align everything on the title page to the left. + headings in the manuscript.
   The title page and headings were all re-aligned to the left.

3. Please indicate the corresponding author with a symbol.
   The corresponding author is now indicated with an asterisk (*).

4. Change Author Contact Information: -> emails:
   “Author Contact Information:” was changed to “Email:” as found in other previously published articles for this journal.

5. Abstract - Please ensure that the Abstract in the submission system matches exactly that in the manuscript file. See submission system for example of how it should look.
   The abstract in the manuscript was copied and pasted into the submission system
6. Please give the tables a section heading i.e. 'Tables'.
   The section heading ‘Tables’ was added.

7. Please remove the blank page from the end of the document.
   The blank page was removed.

8. Reference 23 - The reference list should contain all authors' names, if there are 30 or below, the term 'et al.' should not be used. If there are more than 30 names a minimum of 10 should be listed, and then et al can be used.
   The original abstract for this reference was located and the reference was changed to include all 10 authors listed in the abstract followed by “et al.”